Regional Organizations and International Politics: Trading Asian Development Bank Loans for United Nations Security Council Votes

Daniel Yew Mao Lim and James Raymond Vreeland Georgetown University

Word Count: 12,978

January 2011

Abstract Japan has consistently sought influence over the United Nations Security Council (UNSC), a platform from which it seeks to shape global affairs. As Japan has a privileged position in the governance of another international organization – the Asian Development Bank (ADB) – we investigate whether Japan leverages its power in this financial institution to facilitate project loans for the UNSC's elected members in exchange for their favorable votes. Analyzing panel data of ADB loan disbursements to 24 developing member-countries from 1968-2002, we find evidence that temporary UNSC membership increases a typical country's share of ADB loans by approximately 25 percent.

1. Introduction

Discontent with the political manipulation of global institutions by Western Powers has led some prominent statesmen to advocate the regionalization of global governance. In the immediate aftermath of the Cold War, Malaysian Prime Minister Mahathir Mohammed urged the Association of Southeast Asian Nations (ASEAN) to form a Japan-led East Asian Economic Group (Rozman 2007). He argued that a regional arrangement would afford the small countries of ASEAN "a bigger say" (Severino 2007: 6). Mahathir's proposal was ahead of its time and received lukewarm responses from Japan and most Asian countries, but since the East Asian Financial Crisis, Japan has successfully spearheaded, and most Asian countries have favored, greater coordination through regional organizations.¹ This significant about-turn in collective regionalist sentiment finds reflection in the formal establishment of the Chiang Mai Initiative, a major step towards the development of an Asian alternative to the IMF. Asia represents but one example of a global proliferation in regional financial institutions. In 2009, Latin America established the Banco del Sur as an alternative to the World Bank and IMF; the African Union countries signed an agreement commissioning the headquarters of the African Monetary Fund in Cameroon. In Europe - where regionalization is most advanced - Germany continues to spearhead the creation of a European Monetary Fund.

Some advocates in Latin America and Asia – especially since the East Asian Financial Crisis – contend that the United States' dominance in global institutions allows it to willfully impose its preferred policies, and that regional organizations would better safeguard their interests. Yet, skeptics – some from middle powers like the Republic of Korea – question if a

¹ See Lipscy (2003) for an overview of Japanese initiatives to spearhead greater regional organizations.

move from global to regional governance would simply shift the locus of political manipulation from the global to the regional level. If the world were to embrace regionalism as the new form of global governance, would political manipulation cease greasing the wheels of international institutions? Or would regional organizations, just like their global counterparts, remain subject to political manipulation, promoting the foreign policy preferences of regional, rather than global, hegemons?²

Our study addresses these questions by focusing on political manipulation within the international financial architectural framework. Specifically, we consider political manipulation at the Asian Development Bank (ADB). As an "open" regional organization, the membership of this institution includes Western Powers, particularly the United States, yet it privileges Asian countries. The country most privileged in the governance of the ADB is Japan. We investigate whether Japan leverages its influence to facilitate project loans for countries it considers politically important.

To measure political importance, we use temporary membership on the United Nations Security Council (UNSC), a body in which Japan has invested significant international political capital. Influence on the Security Council has long been a top foreign policy goal for Japan, exhibited by its failed campaigns for a permanent seat and its repeated successful election to the body on a temporary basis. Japan's exclusion from UNSC discussions of major international security issues is generally perceived as incommensurate with its status as an economic power, a reality regarded by many within the Japanese government as a national humiliation. Japan also has intense preferences over the perennially discussed issues concerning North Korea.

 $^{^{2}}$ For the seminal piece on the political manipulation of the IMF, see Thacker (1999). For work on the manipulation of the World Bank, see Fleck and Kilby (2006), Kilby (2008, 2009).

The literature on the trading of UNSC votes for foreign aid focuses on the United States and its political manipulation of international organizations. We build on and make two explicit contributions to this literature. First, we explicate the causal mechanism of global horse-trading through international financial institutions, adopting a principal-agent framework and disaggregating the state into its Ministry of Finance and Ministry of Foreign Affairs. Second, we present the first systematic evidence of global horse-trading by a regional hegemon at the regional level.

Our argument, then, is that Japan engages in global-horse trading through the ADB, exchanging ADB loans for countries' favorable votes at the UNSC. Note that Japan and the United States – the second most powerful country at the ADB – are close allies and hardly ever oppose one another on the UNSC. We therefore suspect that it is easy for them to agree on the political importance of temporary UNSC members.

As for whether Japan takes the lead, previous studies that focus on the US-Japan relationship dynamic have found evidence that Japanese foreign policy is generally formulated as a reaction to US pressure (*gaiatsu*) (see Miyashita 1999, Drifte 2000).³ For example, Japan had "yielded to the United States" in withholding its support for the formation of the East Asian Economic Grouping and abandoning the idea of establishing an Asian Monetary Fund. One could therefore argue that Japan's close alignment with the United States on the UNSC reflects *gaiatsu*,

³ Drifte (2000: 2-3) provides an excellent summary of Japanese reactive state arguments:

[&]quot;multilateral diplomacy has been subordinated to American foreign and security policies and developed in a rather reactive way...it is still lacking in political ideas, concepts, and personnel input."

and consequently interpret political manipulation at the ADB as resulting from US, rather than Japanese, efforts.

Still, previous studies (Kilby 2006, 2010), our analysis of the formal control of the ADB, and evidence from our interview with a former Director-General at the institution all point to the conclusion that American influence at the ADB goes through channels controlled by Japan, the senior partner at this international organization.⁴ Moreover, recently Japan has spearheaded several initiatives at the United Nations (UN) without explicit American prodding.⁵

Our main conclusion is therefore that the ADB – a regional organization that Japan dominates – is subject to political manipulation, and its lending practices favor countries that are strategically important to Japan. We rely on both qualitative and quantitative evidence to support our claims. Qualitatively, an interview with a former Director-General at the ADB suggests that the ADB is de facto "funded by the Japanese, controlled by the Japanese, and run by the Japanese" (Bello 2000). Primary and secondary historical sources corroborate this interpretation. The key contribution of our study, however, is quantitative. We consider a panel of ADB loan shares disbursed to 24 developing member-countries from 1968-2002. We begin with a propensity score matching approach that involves pairing countries that served on the UNSC with comparable countries that did not, and then comparing the average difference in ADB loan shares. Next, we use standard regression analyses, accounting for country and year fixed effects and auto-correlated error terms as well as a range of control variables suggested by the seminal

⁴ 17 August 2010, hereafter referenced as "interview."

⁵ These include the streamlining UN administrative processes, the initial proposal for an international arms trade treaty (Erfat 2010), and the dispatch of UN peacekeeping troops to Cambodia in 1992 (Dore 1997).

study of Kilby (2006). We find that temporary UNSC membership increases a typical country's share of ADB loan disbursements by approximately two percentage points. Given that the average ADB loan share is about 7.5 percent, this represents a substantial increase for the typical country. Robustness tests further suggests that it is indeed Japanese influence that accounts for the effect of UNSC membership on ADB lending.

The rest of the paper proceeds as follows. Section 2 provides an overview of the ADB as well as evidence of the politics behind ADB lending gathered from an interview with a former ADB Director-General. Section 3 presents a brief background on the UNSC. Section 4 outlines our argument, while section 5 details our empirical results. Section 6 concludes.

2. Japan and the ADB

The ADB was founded on December 19, 1966 as an Asia-Pacific version of the World Bank.⁶ Unlike the "closed" regional organizations that have been emerging in Asia that have exclusively Asian membership, the ADB is an "open" regional organization with prominent Western members.⁷ Still, its governance privileges Asian countries, especially Japan.

⁶ For an overview of the historical developments leading up to the ADB's establishment in 1966, see Kappagoda (1995). For an overview of the politics of ADB lending (as well as those of other regional development banks), see Strand (2011).

⁷ For an overview of "open" as opposed to "closed" regional organizations, see Bulmer-Thomas (1998). For work on the closed organization of ASEAN Plus Three, see Rozman (2007). For work on the Chiang Mai Initiative, see Lipscy (2003), Eichengreen (2003), Henning (2009), and Grimes (2010a,b).

The Board of Governors, which includes representatives from all member-countries, sits at the apex of the ADB hierarchy and delegates operational responsibilities to a smaller Executive Board of Directors.⁸ Voting power is related to each member's share of capital subscription to the ADB, which, in turn, is related to the economic size of the country and whether it is located in Asia. Asian countries contribute a significantly higher proportion to the ADB's subscription share (63 percent) than they do, for example, at the World Bank (22 percent), in return for a greater share of the votes (65 percent vs. 22 percent).⁹

The United States and Japan hold equivalent voting power (12.8 percent each), and are by far the ADB's most influential members (Kilby 2006). Japanese and American nationals occupy many of the ADB's senior management posts, with the most powerful positions going to Japan. The Director-General heading the Budget, Personnel, and Management Systems Department is the second most powerful person at the ADB and has always been from Japan.¹⁰ Most importantly, all eight of the Presidents of the ADB have come from Japan, with seven of them coming directly from the Ministry of Finance (MOF).

As a former Director-General at the ADB explains: "the ADB is like a state, and the ADB President is like its emperor."¹¹ The President of the ADB acts as the Chairman of the Executive Board and wields enormous influence. According to the voting procedure, the President may, in the absence of a request for a formal vote, "ascertain and announce to the meeting the sense of

⁸ http://www.adb.org/About/management-staffing.asp

⁹Hhttp://siteresources.worldbank.org/BODINT/Resources/278027-

^{1215524804501/}IBRDCountryVotingTable.pdfH, http://www.adb.org/About/membership.asp

¹⁰ Interview, 17 August 2010. Also see Kappagoda (1995).

¹¹ Interview, 17 August 2010.

the meeting with regard to any matter and the board shall be deemed in accordance with the announcement by the Chairman without the necessity of taking a formal vote."¹² Put differently, the ADB endows its President with agenda-setting power; he alone determines which projects are brought to the Board of Directors for a vote. Should the occasion for a formal vote arise, the ADB President can make proposals that effectively function as "no-action default" policies, making him the "conditional agenda setter."¹³ If an Executive Board Director wishes to oppose a proposal made by the President, he must undertake a series of costly actions, such as requesting a formal vote and putting together a coalitional majority to win the vote. Collective action problems and psychological inertia render this task an uphill battle (Korobkin 1998, Camerer et al. 2003, Thaler and Sunstein 2008).

So, whose interests does the President represent? Officially, he reports to the Chairman of the Board of Governors. Yet he appears to make most of his decisions in consultation with the Japanese MOF, which acts through the Japanese representative on the Board of Directors.¹⁴ Now, the ADB Presidency is a sunset post for officials from the Ministry of Finance – the mandatory retirement age for Japanese civil servants is 60, and many ADB Presidents were appointed in their late fifties. From a principal-agent perspective, this raises potential concerns about the extent to which the ADB President's preferences could diverge from the Japanese MOF's. We suspect that agency slack is minimized for two reasons. First, potential candidates for the ADB

¹² See the Asian Development Bank Board Rules, Section 5(a), pp2-3. Available: Hhttp://www.adb.org/BOD/Rules_Procedure_Brd_Dir_ADB.pdfH.

¹³ See Tsebelis (1994), Tsebelis and Kreppel (1998), Stone (2011), Ayres and Gerner (1989), Johnson and Goldstein (2003). Also see the ADB *Articles of Agreement*, Article 33(3).

¹⁴ Interview, 17 August 2010.

Presidency are drawn from a small pool of candidates with significant ties to the MOF. The MOF is thus aware of their preferences, and would likely appoint agents with similar preferences to its own ex-ante. Second, the MOF reserves the right to remove an ADB President ex-post. For example, Masuo Fujioka (1981-1989) was replaced in part because his authoritarian style; his successor, Kimimasa Tarumizu (1989-1993), was replaced because of his poor leadership.¹⁵

Thus the dynamics of the Executive Board of Directors appear to accord the Japanese government – acting mainly through the ADB President – tremendous influence over the loan approval process. A lot of backroom dealing goes on, and all projects require Japanese support; otherwise, they do not go through to the Board of Directors for approval. Interestingly, even when the United States and some other countries occasionally object to a particular loan, it may be approved if it has Japanese backing.¹⁶

In terms of systematic evidence, the only other quantitative studies on the political manipulation of the ADB (of which we are aware at the time of writing), show that countries deemed important to Japan and the United States receive favorable treatment from the ADB (Kilby 2006, 2010). Specifically, countries receiving a greater share of bilateral foreign aid from Japan are more likely to receive loans from the ADB. They also receive a larger allocation of total ADB lending. Countries receiving a greater share of bilateral foreign aid from the United States are also more likely to receive loans from the ADB, although the size of the allocation appears unaffected. Kilby tests other measures of importance – vote alignment with the United States and Japan at the United Nations General Assembly and the strength of trade ties – but the results are either not robustly significant across all specifications or do not have the expected

¹⁵ New York Times (2 July 1993) "ADB Chief Quits Amid Mounting Criticism of Role."

¹⁶ Interview, 17 August 2010.

signs. These ambiguities, however, do not negate the credibility of the findings on bilateral aid, particularly those that demonstrate "the tighter link (when compared to the United States) between ADB lending and Japanese aid" (Kilby 2006: 192). In short, Kilby's findings suggest that it may be Japanese, rather than American, influence that explains the international politics of ADB lending.

The connection that Kilby establishes between ADB lending and Japanese foreign aid is convincing, though it remains possible that both are driven by other concerns. Tuman et al. (2009), for example, find little evidence that Japanese foreign aid is driven by economic or security interests; they instead find that it is driven by humanitarian concerns. It is thus conceivable that the correlation between Japanese and ADB aid is driven by economic, not political, factors. That said, we believe that Kilby's interpretation is correct. As argued by Morgenthau, "the transfer of money and services from one government to another performs here the function of a price paid for political services rendered or to be rendered" (1962: 302). The veracity of this interpretation would be even more convincing if a more exogenous measure of political importance could be employed.

Japan has consistently displayed interest in the Security Council; one need only look to its failed attempts to gain a permanent seat (Weiss 2008) and its successful campaigns to win temporary membership. We suspect that Japan also seeks to extend its influence over the UNSC through the other elected members. Recent work corroborates our hypothesis – Frankel and Gibbons (2010) find that a country's temporary UNSC membership status has a positive and statistically significant impact on the share of Japanese Official Development Assistance received. We investigate whether Japan might also wield its influence over aid flows to developing countries at the ADB to further its foreign policy objectives at the UNSC.

3. Japan and the United Nations Security Council

The United Nations (UN) was established in 1945 by the victors of World War II with the maintenance of international peace and security as its primary raison d'être.¹⁷ Within the UN architecture, the UNSC is the most powerful organ. Article 39 of Chapter VII of the UN Charter grants the UNSC the power to determine if there has been a threat or breach to the peace, or if an act of aggression has taken place. Articles 41 and 42 permit the UNSC to apply economic and diplomatic sanctions and to authorize the use of military force.

Today's UNSC comprises five permanent members and ten non-permanent members. The composition of the permanent members has remained unchanged since the UN's founding – China, France, the Soviet Union (now Russia), the United Kingdom, and the United States. These five countries possess veto power over UNSC resolutions. The ten non-permanent members represent designated regions and are selected in staggered elections held every October for two-year (limited) terms beginning January 1 the following year. A two-thirds majority vote is required from the UN General Assembly to win election. Since 1966, there are three seats for Africa and two for Asia (with at least one of these five seats going to an Arab country), one for Eastern Europe, two for Latin America, and two for Western Europe.¹⁸ Winning election to the UNSC is no accident – Malone (2004) reports that there is extensive competition for seats. Still, patterns of election appear to be idiosyncratic and term limits assure the exogeneity of exit (Dreher and Vreeland 2010).

¹⁷ See the UN Charter, Article 1.

¹⁸ The Western Europe group includes "others," like Canada, Australia, and, recently, Turkey. For more on the UNSC, see Russett (1997).

How important are these elected UNSC members? Article 27 of the UN Charter stipulates that absent a veto from a permanent member, nine out of fifteen votes are required for a Security Council resolution to pass. This means that at least four out of the ten elected members must vote in favor of a resolution for it to pass, more if any of the permanent members abstain. Because they lack veto power, O'Neill (1996) finds that non-permanent members control less than two percent of total voting power.¹⁹ Yet, the political importance of each non-permanent member may derive from the legitimacy that its votes can confer (Voeten 2001, 2005, Chapman 2007, Hurd and Cronin 2008, Chapman forthcoming). As Hurd (2007) explains, the elected members serve the purpose of giving voice to the "rest of the world" on the Security Council. Legitimacy may derive from the representational role that the elected members play, or from the informed nature of their decisions. Members of the UNSC have access to sensitive documents and private discussions regarding the importance of taking international action.²⁰ And the legitimizing effect of the Security Council may extend beyond the international level into domestic politics (see Chapman and Reiter 2004).

As a defeated power in World War II, Japan was excluded from the United Nations until 1956.²¹ It therefore comes as no surprise that it was not awarded a permanent seat on the Security Council by the victorious powers. A global power before and during the war, Japan rose again in

¹⁹ Also see Strand and Rapkin (2010). Note that this ignores the possibility that oversized coalitions may be sought. On the logic of buying extra votes, see Volden and Carrubba (2004). For a contrasting view, see Saiegh (2011).

²⁰ For more on the informational story, see Fang (2008), Chapman (2007), and Thompson (2006).

²¹ Japan gained admission on 18 December 1956. Article 53 of the UN Charter hints at the bias held by the Allies against the Axis powers.

the decades that followed. Japan grew to become a top five economy by the time it regained its independence in 1952; it was the second largest economy in the world by 1964.²² It has since sought a position on the global stage equivalent to its economic standing. Japan's aspirations of becoming a true global power are, however, constrained by Article 9 of its constitution, which renounces "war as sovereign right of the nation and the threat or use of force as a means of settling international disputes" (Dore 1997: 106). Article 9 further specifies that Japan will never maintain "land, sea and air forces, as well as other war potential" (Dore 1997: 106). To have a presence in matters of global security, Japan can act legitimately only through the UN Security Council. The Yoshida Doctrine's emphasis of Japan's UN-centered foreign policy (alongside US bilateralism) reflects this general desire. Indeed, the Japanese MOFA has consistently sought an important role for Japan in the institution. Japan first served on the UNSC in 1958-1959, two years after gaining accession to the United Nations. Shortly after serving its second term in 1966-1967, the MOFA revealed its desire for Japan to be "regarded as a major UN member state" by announcing that it would seek election to the UNSC "as often as possible" (Dore 1997: 29). While it has waged unsuccessful campaigns to obtain a permanent seat (Heenan 1998, Weiss 2008), it is arguably the most successful in winning election. Since its accession to the United Nations, it has been elected to serve a total of ten times, more so than any other country in the world save Brazil (with which it is tied). Winning election to the UNSC is no accident – Malone (2004) reports that there is extensive competition for seats. Given Japan's interest in participation, we suspect that it also seeks to extend its influence over the UNSC through the other elected members.

 ²² Source: Penn World Table 6.2 – Real GDP per capita (Constant Prices: Chain series) ×
Population.

4. A Theory of Global Horse-Trading

Overview

Trades of political influence for money are possible because governments sometimes have divergent preferences. Though developing countries that are elected to the UNSC for a two-year term have the opportunity to vote on globally important issues, they typically care more about loans than they do about most UNSC votes. In contrast, the hegemonic major vote-holders at the International Financial Institutions (IFIs) often care more about the activities of the UNSC. Governments can therefore maximize their respective preference functions by seeking out trades. ²³

Even so, why would the hegemon choose to employ an IFI as a vehicle for securing political influence in the UNSC when it can achieve the same outcome through the use of bilateral aid? We identify three advantages that an IFI may offer.

The first involves obfuscation. As suggested by economists like Vaubel (1986) and political scientists like Abbott and Snidal (1998), international organizations can be used to "launder" the "dirty work" of governments. Hegemons typically have a checkered history of imperialism, and often try to avoid projecting a neo-imperialist image in their dealings with developing countries. Trading loans for votes through an IFI allows them to better mask their influence.

²³ Political Scientists have shown that these trades might be detrimental to developing countries in the long-run. Bueno de Mesquita and Smith (2010) find that countries that win election to the UNSC subsequently experience a substantial decrease in their economic and democratic performance.

The second centers on cost sharing (Dreher 2009a,b). Rather than provide individual assistance, the major vote-holders can take advantage of the common resource pool that the IFI offers. They can also leverage the IFI's infrastructure to more effectively raise funds on international capital markets, disburse loans, and manage repayments. All these reduce the cost of issuing loans from the hegemon's perspective.

The IFI's infrastructure offers a third advantage: technical assistance. Aid from the IFI does not just involve lending. It also involves policy advice as well as assistance in identifying, developing, and implementing a project. For example, developing countries that wish to develop their financial infrastructure might lack the requisite technical knowledge to undertake the optimal sequence of reforms that will allow them to do so. IFIs have often accumulated extensive expertise on these and other types of development challenges. They can therefore more effectively assist these developing countries compared to donor countries that simply offer a lump sum of money.

Articles published in economics journals provide evidence that trades of IFI loans in return for UNSC votes take place. The United States appears to use its influence in IFIs to increase aid flows to elected UNSC members. Temporary UNSC members have been found to receive more UN Development Program aid through UNICEF than non-members (Kuziemko and Werker 2006). They also enjoy more project-loans from the World Bank, and more IMF loans with fewer conditions attached from the IMF (Dreher et al. 2009a,b, 2010).

Though these articles establish a statistically significant correlation between UNSC membership and IFI loans, they do not explicate the causal mechanism that links UNSC membership to IFI loans. How exactly does a country leverage its influence at an IFI to increase its influence at the UNSC? Does the negotiation required to facilitate global horse-trading occur

15

primarily at the international or domestic level? We present a theory of global horse-trading that focuses on domestic bargaining.²⁴

Actors and Preferences

We consider the state as disaggregated into its Ministry of Finance (MOF) and its Ministry of Foreign Affairs (MOFA), and focus on the international dimensions of their operations.²⁵ The hegemon's representatives at an IFI are the agents of the MOF, whereas those at the UNSC are the agents of the MOFA. Each ministry selects a policy option to maximize its international objective function. The MOF chooses a loan policy option to maximize expected returns on its IFI capital contributions. The MOFA chooses a foreign policy option to maximize the hegemon's national interests. We focus on the multilateral aspect of the MOFA's foreign policy options, specifically those that involve international institutions such as the UNSC. The MOFA often seeks to attain a unanimous vote for UNSC resolutions that concern its national interests. Maximizing its national interests thus comprises two sequential steps – (1) choosing how to vote on a specific UNSC resolution, and (2) persuading other members on the UNSC to cast a similar vote. We examine how the MOFA may use IFI loans to win favor with temporary UNSC members.

²⁴ For a domestic approach that looks at US foreign aid policies, see Milner and Tingley (2010a,b).

²⁵ Our motivation for doing so derives from Liberal IR theory, which conceives of states as comprising groups that possess heterogeneous preferences (see, e.g., Moravcsik 1997, Lim 2010).

Mechanism

The functional specialization of government bureaucracies implies that any attempt to use the IFI as a vehicle through which to engage in global horse-trading involves bargaining among the relevant government ministries. Specifically, to maximize its international objective interests at the UNSC, the MOFA would need to convince the MOF to issue a loan through an IFI to the UNSC member whose vote it sought. The MOF's and the MOFA's interests, however, may conflict – the provision of loans based on political rather than economic criteria may decrease the expected returns on IFI capital contributions. The mechanism that governs global horse-trading is therefore a circuitous one that requires the MOFA to expend political capital to get the MOF to facilitate favorable loans for elected UNSC members. Because of the costs involved, the MOFA will only seek to do so when it has a specific agenda.

A reverse logic applies from the perspective of a developing member-country – its MOF must expend political capital to get its MOFA to vote accordingly at the UNSC. Since a typical country rarely serves on the UNSC, the MOF will seek to capitalize on the one-off windfall.

Should a significant issue come up during the tenure of such a temporary UNSC member, it behooves Japan to have the country in its debt, and ADB loans are a low-cost way to achieve this goal. Thus, we suspect that the representatives of the MOF are made aware by the MOFA of the potential importance of developing countries serving on the UNSC. If temporary UNSC members request a loan, the Executive Board and the Management facilitate approval, with the perhaps vague implication that the loans will be cut off if they misbehave on the UNSC. It is best to put the temporary UNSC members in a position of owing Japan favors.²⁶

²⁶ The logic of the incentive approach to gaining policy favors is explored by Dorussen (2001).

We thus formulate two hypotheses to test in the subsequent sections: (1) Temporary UNSC members receive a greater share of ADB loan disbursements than non-members; (2) It is Japanese, rather than American, political manipulation that accounts for the effect of UNSC membership on ADB lending.

5. Quantitative Evidence

In this section, we present quantitative evidence supporting our two hypotheses. We begin by describing the data and presenting descriptive evidence of our broad hypothesis that UNSC members receive a greater share of ADB lending. We then test the hypothesis using more rigorous methods. Finally, we present suggestive evidence in favor of our second hypothesis, that it is Japanese political manipulation of the ADB that accounts for the effect of UNSC membership.

Note that the initiation of the exchange of favors may be idiosyncratic. Just as Japan may use its leverage at the ADB to gain influence over the UNSC, elected members of the UNSC may use their leverage to gain influence over the ADB. The quantitative approach we adopt in this paper certainly does not allow us to distinguish between who approaches whom, and we recognize that both avenues are possible. Our goal is to establish that there is a robustly significant correlation between UNSC membership and ADB lending.

5.1 The share of ADB lending and UNSC participation

Our dataset comes from Kilby (2006) and covers 24 developing member-countries from 1968 to 2002 for a total of 467 country-year observations. The panel is unbalanced due to varying dates of country independence and eligibility for ADB lending. The share of ADB loan disbursements

that countries receive varies from 0.0001 to 0.52. The mean allocation is 0.07, and the median is 0.04. All eligible countries receive some funding every year.²⁷

Turning to our independent variable of interest, Table 1 lists the countries in Asia that have served (or not) on the UNSC. Only nine of the twenty-four countries in our sample serve on the UNSC between 1968 and 2002: Bangladesh, India, Indonesia, Nepal, Pakistan, South Korea, Malaysia, Philippines, and Thailand.²⁸ Of these countries, most have served multiple terms on the UNSC, with India and Pakistan serving five and four terms, respectively. South Korea and Thailand have each served only once.

Does temporary membership UNSC membership influence ADB lending? Figure 1 presents descriptive evidence suggesting that it does. Consider first the left-hand part of the figure. The average share of annual lending that goes to countries not serving on the UNSC is about 0.07, while the average share going to elected members of the UNSC is 0.12. A t-test indicates that this difference is significant at the 0.01 level (t-statistic=3.01). The two-sample Kolmogorov-Smirnov (KS) test also rejects the null hypothesis of equality between the empirical

²⁷ Note that we use data on loan disbursements not commitments because the former are more readily available, although results might be crisper using commitment data. New research suggests that international politics can have direct influence on ADB commitments and indirect and informal influence through the ADB staff, who actually disburse the loans (Kilby 2010). This may influence the timing of disbursements, which we address below.

²⁸ Singapore served on the UNSC from 2001-2002, but we have excluded it from our analysis because we only have ADB loan data for Singapore until 1997. As for China, as it is a permanent member of the UNSC, we do not code our UNSC variable 1. When we do, however, the main results hold (results available on request).

distribution functions of ADB loan share for UNSC members and non-members at standard levels of significance (p-value=0.002).

More convincing is the pattern over time, presented on the right-hand part of Figure 1. Two years before election to the UNSC, the average share of annual ADB lending is similar to that of non-member-countries: about 0.08. The share is slightly lower the year before election: 0.07. In the election year, the share goes up to about 0.10. During the first and second years of membership on the UNSC, the share of annual ADB lending rises to 0.11 and 0.12. Then exogenous term limits force members to step down. The share of lending drops back to 0.10 and 0.09 in the first and second years after the UNSC term ends. It remains constant at about 0.09 in the third year after before returning to almost pre-UNSC levels of 0.08 in the fourth year.

The peak in the share of ADB loans received during the second year of UNSC membership appears to be driven by the ADB's project lending cycle, which is illustrated in Figure 2.²⁹ The ADB's project lending cycle comprises five main stages: (1) Country Strategy Partnership, (2) Preparation, (3) Appraisal/Approval, (4) Implementation, and (5) Evaluation. Stages 2 and 3 concern the time required to put together a project, and hence the most relevant to our analysis.

The average duration from project proposal to disbursement of ADB loans ranges from 1.5 to 2.5 years. If the ADB expedites the project loan process after a country is elected to the UNSC (three months before its term begins), then the average time of 18 months before ADB loans are disbursed accounts for the peak in ADB loan shares during the second year of

²⁹ See Hhttp://www.adb.org/projects/cycle.aspH for a general overview of the ADB Project Cycle. Our Director-General interview confirms this. For details, see the "Business Processes For The Reorganized ADB" (2001).

membership on the UNSC. This schedule helps to explain both the peak in the second year as well as the lag before lending returns to levels prior to UNSC membership. Note that from a game-theoretic point of view, it is important that the ADB (or Japan and the United States through the ADB) move last when it comes to the trade of loans for votes. For the elected member of the UNSC, the two-year term is essentially a one-shot game, since governments serve rarely. Japan and the United States play this game repeatedly with each new UNSC member, and thus have a reputation at stake.

Country	Served on UNSC	Eligible for ADB lending	Non- UNSC Ave. ADB share	UNSC Ave. ADB share	Fits hypo- thesis?
Japan	1958-9, 1966-7, 1971-2, 1975-6, 1981-2, 1987-8, 1992-3, 1997-8, 2005-06	No	n/a	n/a	n/a
India	1950-1, 1967-8, 1972-3, 1977-8, 1984-5, 1991-2	After 1987	0.11	0.15	Yes
Pakistan	1952-3, 1968-9, 1976-7, 1983-4, 1993-4, 2003-04	Yes	0.14	0.15	Yes
Philippines	1957, 1963, 1980-1, 2004-05	Yes	0.10	0.18	Yes
Bangladesh	1979-80, 2000-01	Yes	0.078	0.084	Yes
Thailand	1985-6	Yes	0.09	0.12	Yes
S.Korea	1996-7	Yes	0.14	0.16	Yes
Malaysia	1965, 1989-90, 1999-2000	Yes	0.06	0.02	No
Nepal	1969-70, 1988-9	Yes	0.02	0.02	No
Indonesia	1973-4, 1995-6, 2007-08	Yes	0.16	0.15	No
Syria	1947-8, 1970-1, 2002–03	No	n/a	n/a	n/a
Iraq	1957-8, 1974-5	No	n/a	n/a	n/a
Jordan	1965-6, 1982-3	No	n/a	n/a	n/a
Turkey	1951-2, 1954-5, 1961	No	n/a	n/a	n/a
Bahrain	1998-9	No	n/a	n/a	n/a
Iran	1955-6	No	n/a	n/a	n/a
Kuwait	1978-9	No	n/a	n/a	n/a
Lebanon	1953-4	No	n/a	n/a	n/a
Oman	1994-5	No	n/a	n/a	n/a
Qatar	2006-07	No	n/a	n/a	n/a
Singapore	2001-02	Yes	n/a	n/a	n/a
Sri Lanka	1960-1	Yes	n/a	n/a	n/a
U.A.Emirates	1986-7	No	n/a	n/a	n/a
Yemen	1990-1	No	n/a	n/a	n/a
Vietnam	2008-09	Yes	n/a	n/a	n/a

Table 1: UNSC participation for the Asian Group

Never members (for those eligible for ADB lending, the average ADB share is 0.01; those not eligible are marked with *): Afghanistan, Armenia, Azerbaijan, Bhutan, Brunei Darussalam, Cambodia, Cyprus*, North Korea*, Georgia, Israel*, Kazakhstan, Kiribati, Kyrgyzstan, Laos, Maldives, Marshall Islands, Micronesia, Mongolia, Myanmar, Nauru, Palau, Papua New Guinea, Samoa, Saudi Arabia*, Solomon Islands, Tajikistan, Timor-Leste, Tonga, Turkmenistan, Tuvalu, Uzbekistan, Vanuatu.

Sources: http://www.un.org/sc/members.asp, http://www.adb.org/Documents/Manuals/Operations/OMA1.pdf



Figure 1: Allocation of ADB lending by UNSC membership over time

Note: The horizontal line shows the mean share of ADB lending across the entire sample (0.075, n=467).

Figure 2: ADB Project Lending Cycle



Note: From initiation until disbursement takes approximately 1.5 to 2.5 years.

5.2 The selection problem

Table 1 raises an obvious question: Who gets elected to the UNSC? Considering that large and strategically important countries like Japan, India, and Pakistan have been elected the most often, one might ask if perhaps important countries are both more likely to serve on the UNSC and to receive a larger share of loans from the ADB. If so, our analysis may suffer from a problem of non-random selection.

There are several reasons to believe that it does not. First, even if important countries are more likely to be selected, the United Nations observes strict two-year term limits so that the termination of UNSC participation is exogenously enforced. This also means that while reelection is influenced by a country's size, the timing of reelection remains idiosyncratic. Additionally, we control below for population in our analysis of ADB lending, and like Kilby, find that larger countries do not receive the largest allocation – instead, the effect of population follows an inverted U-shape for the share of ADB lending. The selection of populous countries onto the UNSC thus should not be reflected in more loans from the ADB.

Still, to address the possibility of a selection problem, we consider a propensity score matching approach suggested by Rosenbaum and Rubin (1983), using a Gaussian-Kernel

estimator suggested by Heckman et al. (1998).³⁰ Implementing the Gaussian-Kernel matching estimator consists of two steps.³¹

We first calculate the propensity score of UNSC membership. Estimating the propensity scores is not straightforward because country-year observations of UNSC and non-UNSC members are not independent. According to UN rules, only two countries may represent the Asian region each year. We thus use a conditional logit model to estimate the probability of UNSC membership conditioned on the fact that only two countries can be selected each year.

As determinants of UNSC membership, we include population share (proxy for country size) and the average monthly manpower provided by countries to UN peacekeeping missions (proxy for UN General Assembly norm of choosing countries committed to peace, from Heldt and Wallensteen 2006).³² Some argue that only reliable predictors should be employed.³³ We

³¹ We use the psmatch2 program by Leuven and Sianesi (2003). See Nichols (2007) for an excellent overview of causal inference in Stata. For an approach that explicitly combines matching with regression analysis see Imai and van Dyk (2004). Also see Smith and Todd (2005) and Dehejia (2005). For another method, see Ho, Kosuke, King, and Stuart (2007).

³² Our choice of control variables for UNSC selection follows Dreher and Vreeland's (2010) study. Like them, we find selection onto the UNSC to be rather idiosyncratic. At least we find that prior share of ADB loans does not predict UNSC selection (See 2a, Column 1).

³³ See Hopkins and Simmons (2005) and Ho, Kosuke, King, and Stuart (2007).

³⁰ Following Bueno de Mesquita and Smith (2010), we initially tried a nearest-neighbor matching model (Abadie et al. 2001), but we only had one match (compared to Bueno de Mesquita and Smith's four). Furthermore, we decided to use propensity score matching since matching suffers from the well-documented dimensionality problem (Heckman et al. 1998).

find the effect of both controls to be positive and statistically significant at the 0.01 level (see Table 2a, column 2). We also tested other potential determinants including: a variable measuring length of time since last UNSC service, IMF program participation, World Bank programs, share of US loans, share of US grants, per capita income, civil war, international war, debt service, US voting in UN General Assembly, USSR/Russia voting in General Assembly, percent of the population that is Muslim, percent Catholic, a measure of shared regional ideology, British colonial legacy, and a Democracy indicator. None of these robustly predicts UNSC membership for the Asian region.

Next, we match each UNSC member against a non-UNSC member with a similar propensity score, and obtain 29 matches.³⁴ We then estimate the effect of UNSC membership on share of ADB loans received – the average effect of the treatment on the treated – by taking the

³⁴ Our sample contained 30 treated observations and 437 untreated observations. We applied the Gaussian-Kernel estimator and selected a bandwidth parameter of 0.025. In other words, we defined the subset of control observations for each treated observation to be those untreated observations with a propensity score that fall within a 0.025 interval of the treated observation's propensity score. Consequently, we obtain 29 treated observations and 325 control observations. Each of these 29 treated observations was matched to at least one control observation, though Stata does not allow us to distinguish how many controls were matched to each of the 29 treated observations. One could say that we obtained 29 "composite" matches. Note that all the matches fulfill the common support assumption. Visual inspection of the distribution of the treated and untreated observations used in the analysis confirms that the sample is appropriately balanced. The figure, not included due to space constraints, is available on request and with the replication materials.

average difference across these matched observations. Using the parsimonious model of UNSC selection, we find that UNSC membership increases the share of ADB lending by about 0.032 (a relatively large effect considering the mean share is 0.075), an effect that is statistically significant at the 0.05 level (t-statistic=2.12). They are presented in Table 2b in bold. Using the full set of predictors, the estimated effect is similar (0.036, t-statistic=2.18). We also report the differences for the "unmatched" samples, which are larger, indicating that failing to account for systematic differences in the samples creates an upward bias for the effect of UNSC membership on ADB loan share.

Table 2c and 2d present diagnostics showing the balance of our samples with respect to the variables used to predict selection (for the parsimonious and full samples, respectively). The differences in means between the treated and control groups are statistically significant for the unmatched sample but statistically insignificant for the matched sample. The result from this balancing test indicates that we have successfully controlled for observable heterogeneity between the treated and untreated observations in our matched sample.

	Significant predictors with	Only significant	
Variable	lagged ADB Share	predictors	Full controls
ADB share (t-1)	1.81		3.937
	(2.19)		(-2.711)
Population share (t-1)	5.02***	5.06***	8.925***
	(1.44)	(1.41)	(-2.088)
Peacekeeping troops (log)	0.20***	0.21***	0.246***
	(0.08)	(0.07)	(-0.091)
British colonial legacy			0.028
			(-0.475)
Hostility level			-2.054**
			(-1.005)
IMF program participation			0.666
			(-0.451)
GDP per capita			0.061
			(-0.065)
One year after UNSC			-16.528
			(-1798.744)
Two years after UNSC			-15.872
			(-1950.61)
Number of observations:	388	401	376
Log likelihood:	-76.97	-77.65	-67.76
Pseudo R2:	0.12	0.12	0.20

Table 2: Results using Gaussian-Kernel propensity score matching estimator

a) Estimating the propensity of UNSC membership (conditional logit, conditioned on year)

b) Comparison of mean ADB share (with and without propensity score matching):

Only significant predictors		Full controls		
Unmatched sample:	Matched sample:	Unmatched sample:	Matched sample:	
0.109	0.109	0.114	0.113	
0.066	0.078	0.066	0.077	
0.044	0.032	0.048	0.036	
0.014 3.21***	0.015 2.12**	0.014 3.47***	0.016 2.18**	
	Unmatched sample: 0.109 0.066 0.044 0.014	Unmatched sample: Matched sample: 0.109 0.109 0.066 0.078 0.044 0.032 0.014 0.015	Unmatched sample: Matched sample: Unmatched sample: 0.109 0.109 0.114 0.066 0.078 0.066 0.044 0.032 0.048 0.014 0.015 0.014	

c) Balancing test for matching variables: Only significant predictors (325 untreated, 29 treated)

		Mean UNSC obs.	Mean non-UNSC	t-statistic	p-value
	Unmatched				
Population share (t-1)	sample:	0.051	0.028	1.96**	0.05
	Matched sample:	0.051	0.044	0.30	0.76
Peacekeeping troops	Unmatched				
(log)	sample:	3.80	2.01	3.26***	0.00
	Matched sample:	3.80	2.52	1.52	0.13

d) Balancing test for matching variables: Full controls (308 untreated, 28 treated – 1 treated off support)

		Mean UNSC obs.	Mean non- UNSC	t-statistic
	Unmatched			
ADB share (t-1)	sample:	0.097	0.068	2.09**
	Matched			
	sample:	0.094	0.077	0.88
	Unmatched			
Population share (t-1)	sample:	0.052	0.028	1.96**
	Matched			
	sample:	0.042	0.05	-0.35
	Unmatched			
Peacekeeping troops (log)	sample:	3.31	1.95	2.38***
	Matched			
	sample:	3.32	2.47	1.01
British colonial legacy	Unmatched			
	sample:	0.429	0.481	-0.53
	Matched			
	sample:	0.407	0.488	-0.58
Hostility level	Unmatched			
	sample:	0.036	0.065	-0.61
	Matched			
	sample:	0.037	0.045	-0.14
IMF program	Unmatched			
participation	sample:	0.464	0.429	0.36
	Matched			
	sample:	0.444	0.421	0.17
GDP per capita	Unmatched			
	sample:	3.58	3.39	0.36
	Matched			
	sample:	3.65	3.26	0.44
One year after UNSC	Unmatched	<u> </u>		
	sample:	0	0.292	-0.92
	Matched			
	sample:	0	0.004	-0.33
Two years after UNSC	Unmatched	0	0.004	1.02
	sample:	0	0.036	-1.02
	Matched	0	0.005	0.26
	sample:	0	0.005	-0.36

5.3 Regression analysis

We now turn to testing our hypothesis using regression analysis. We start with a simple bivariate ordinary least squares (OLS) estimator, regressing shares of ADB loans on UNSC membership. We then introduce fixed effects for year and for the various regions of Asia (East, Central and West, Pacific, Southeast, and South). After that, we introduce fixed effects for both country and year. Finally, we extend the country and year fixed effects specification to correct for first-order autoregressive error components. Our motivation for doing so derives from Kilby (2006), who posited two reasons for expecting autocorrelation among the error terms: (1) the gradual and phased timing of loan disbursements render them susceptible to autocorrelation; (2) institutional inertia generated by annual budgeting and defensive lending (see Devarajan et al. 1999, Birdsall et al. 2002, Marchesi 2003, and Marchesi and Missale 2004).

We include as control variables all of the factors that Kilby (2006) finds to be statistically significant determinants of ADB lending – either for the decision to allocate any funds to a country or for the share of ADB lending going to a country. Descriptive statistics and data sources are reported in the appendix. One of Kilby's strongest findings is that a country's share of Japanese foreign aid positively and significantly predicts the share of ADB loans that it receives. In contrast, the share of US foreign aid does not predict the share of ADB lending, though it does positively influence the binary decision to allocate ADB funds to a country. Since donor countries typically channel bilateral aid to strategically important countries, Kilby's findings provide evidence that the ADB allocates loans along the strategic preferences of Japan and the United States. The share of world exports entering a country is also found to have a positive effect on the binary decision to lend, while the share of world imports has a positive effect on the share of ADB lending. These findings imply that countries important to global

economic activity receive more attention from the ADB. The share of Japanese imports from a country further predicts the decision to lend (share of Japanese exports to a country does not have a statistically significant effect).³⁵ Kilby also finds, strangely, that the share of US exports to the country has a negative effect on the decision to lend (share of US imports has no statistically significant effect).³⁶ Beyond these measures of the strategic and economic importance of ADB loan recipients, Kilby finds that population has an inverted U-shaped effect on ADB loan allocation. He also finds that per capita income has a negative effect – poorer countries receive larger allocations. Democracy does not have a statistically significant impact on share of ADB lending, though it has a positive and statistically significant effect on the binary decision to lend.

Our OLS analysis (table 3, column 3), broadly confirms the effects of these control variables. When we introduce fixed country effects and control for autocorrelation (table 3, columns 4 and 5, respectively), the statistical significance of most of the coefficients drops.

³⁵ Note that bilateral trade flows capture both economic interest and security concerns. Gowa and Mansfield (1993) find that political-military alliances significantly influence bilateral trade intensity, and that this effect is strongest in bipolar systems.

³⁶ We consider imports and exports separately following Kilby's (2006) work, although it may be better to combine imports and exports into one trade variable, following Kilby (2010), because the two components are typically highly correlated. Our core results are unchanged when we do this (results are available in the replication materials).

Additionally, we find that the effect of democracy is surprisingly negative.³⁷ The most robust effect of the control variables – and the most important for our paper – is that of Japanese foreign aid. It is positive and statistically significant throughout, indicating the important of Japanese political influence over the ADB.³⁸

Turning to our independent variable of interest, UNSC membership is positive and significant across all specifications at the 0.10 level of significance or better. Our preferred specification is the most conservative (table 3, column 5), where we control for country and year fixed effects, autocorrelation in the error term, and all control variables. We estimate that the effect of UNSC membership is to increase a country's share of ADB lending by 1.8 percentage points, and we can say with 90 percent confidence that the true effect is between 0.2 and 3.3 percentage points. Considering that the average share of ADB lending is 7.5 percent, our point estimate of 1.8 represents an increase of nearly 25 percent.

³⁷ This effect is not robust, however, to other measures of democracy. When we include the Cheibub et al. (2010) measure of democracy, the coefficient is positive and not statistically significant (results available on request).

³⁸ Kilby (2006) convincingly shows that Japanese aid does not simply proxy for economic need, by including a variable measuring the aid provided by "small donors," that are well known to disburse aid on a need-basis. When we include this variable, our core results are unchanged (results available in the replication materials).

Table 3: The effect of UNSC membership on share of annual ADB lending

UNIC	(1) OLS 0.045***	(2) OLS 0.039***	(3) OLS 0.022*	(4) Fixed country effects 0.021*	(5) Fixed country effects with AR(1) 0.018*
UNSC	(0.015)	(0.014)	(0.011)	(0.011)	(0.010)
Japan aid share			0.322*** (0.064)	0.329*** (0.075)	0.165** (0.074)
US aid share			0.159*** (0.048)	0.052 (0.051)	-0.023 (0.050)
World exports			0.773*** (0.233)	0.507 (0.379)	-0.186 (0.413)
World imports			0.102 (0.214)	-0.077 (0.273)	0.289 (0.427)
Japan imports			-0.148 (0.111)	0.074 (0.143)	-0.235 (0.240)
US exports			-0.343** (0.151)	-0.894*** (0.252)	-0.009 (0.296)
Population			1.308*** (0.226)	4.464 (3.190)	8.300 (5.394)
Population-squared			-3.857*** (0.595)	-11.926** (4.699)	-15.338** (7.635)
GDP/capita			-0.007*** (0.002)	-0.007** (0.003)	-0.007 (0.005)
Polity			-0.0001 (0.001)	-0.002*** (0.001)	-0.002* (0.001)
Constant	0.072*** (0.004)	0.116*** (0.034)	0.172*** (0.033)	0.074 (0.106)	-0.205*** (0.079)
Asian-Region fixed effects	NO	YES	YES	NO	NO
Country fixed effects	NO	NO	NO	YES	YES
Year fixed effects	NO	YES	YES	YES	YES
Number of observations	467	467	467	467	443
Number of countries	n/a	n/a	n/a	24	23
(Adjusted) R2	0.02	0.16	0.48	0.15	0.20

Table 4: The Impact of UNSC membership on Share of ADB Loans Allocated Over Time

	(1) OLS	(2) Fixed country effects with AR(1)	(3) Fixed country effects with AR(1)
UNSC member (2 years before election)	0.012 (0.021)	-0.014 (0.013)	
UNSC member (1 year before election)	0.004 (0.020)	-0.035** (0.015)	-0.023** (0.010)
UNSC member (election year)	0.029 (0.021)	-0.010 (0.016)	
UNSC member (1st year)	0.046** (0.021)	-0.003 (0.016)	
UNSC member (2nd year)	0.051** (0.021)	0.03* (0.016)	0.031*** (0.011)
UNSC member (1 year after)	0.034* (0.020)	0.000 (0.015)	
UNSC member (2 years after)	0.018 (0.020)	-0.014 (0.014)	
Japan aid share		0.196** (0.076)	0.166** (0.072)
US aid share		-0.020 (0.050)	
World exports		-0.134 (0.410)	
World imports		0.217 (0.423)	
Japan imports		-0.213 (0.237)	
US exports		-0.068 (0.294)	
Population		9.398* (5.365)	11.502** (5.368)
Population-squared		-17.099** (7.644)	-17.426** (7.554)
GDP/capita		-0.005 (0.005)	
Polity		-0.002* (0.001)	-0.002* (0.001)
Constant	0.069** (0.004)	-0.248*** (0.079)	-0.341*** (0.072)
Country fixed effects	NO	YES	YES
Number of observations	467	443	443
Number of countries		23	23
(Adjusted) R2	0.016	0.194	0.154

5.4 Participation over time

Next, we extend our analysis following UNSC membership over time, as depicted in Figure 1. We include separate indicator-variables for: the first and second years before UNSC election, the year of UNSC election, the first and second years of actual UNSC membership, and the first and second years following UNSC membership. First we analyze the effects of these variables by themselves in an OLS model, then we introduce country and year fixed effects, our control variables, and control for autocorrelation of the error term. Finally, we analyze this model again, dropping the variables that are not statistically significant at conventional levels. Table 4 presents the results.

The pattern that emerges from this analysis broadly confirms the pattern in Figure 1, as well as our understanding of the ADB project lending cycle, which we discussed in section 5.1. The most robust effect is that of the second year of UNSC membership. We suspect that this is because of the time required to prepare an ADB project. Note that in the more efficient model, where we drop the independent variables that are not statistically significant at conventional levels, the effect of the second year of UNSC membership is statistically significant at the 0.05 level. The point estimate of the effect is 3.2 percentage points, and we can say with 95 percent confidence that the effect is between 0.7 and 5.6 percentage points.

Of further interest is the statistically significant effect that we detect one year before election to the UNSC. This effect is unexpected, and we find it strange. Countries that eventually get elected to the UNSC receive smaller shares of ADB loans in the year before they run for election. We doubt that the ADB anticipates which countries will be elected and systematically lowers their share of lending. Rather, we suspect that countries suffering from a lack of international attention may be systematically more likely to run for UNSC election – that said,
we did not detect such an effect in our analysis of the determinants of UNSC selection (Table 2 above). We are reassured by this finding, however, that it is not countries generally preferred by the ADB that are elected to the UNSC. If this were the case, we would not expect differences in the coefficients of UNSC membership over time – and we certainly would not find negative effects, as we do for the year before election.

5.5 Japan

Does the United States or Japan exert greater political influence at the ADB? Note that the United States and Japan may tend to agree on the political importance of UNSC members and thus coordinate their political pressure. In a cursory review of 274 resolutions passed by the UNSC when Japan served with ADB-eligible countries, the United States and Japan never voted against each other, although Japan voted for four resolutions where the United States abstained.

Still, it is striking that while the effect of Japanese bilateral aid on ADB lending is so robust, we do not detect a similar effect for US bilateral aid on ADB lending (also see Kilby 2006). As explained in Section 4, the United States is the top vote-holder at more powerful international institutions – the IMF and the World Bank – and thus may expend its efforts there, while Japan enjoys supremacy only at the ADB. As the ADB President also comes from the Japanese Ministry of Finance (MOF), Japan is in a particularly strong position to exert influence. Recall that the causal mechanism we propose is circuitous: to win favor with countries on the UNSC, the Japanese MOFA must petition the Japanese MOF to use its power at the ADB. The MOFA may only be willing to expend such political capital when it has an active UNSC agenda – that is, when its representatives currently serve on the UNSC.

To test this, we decompose our independent variable of interest into two separate variables: UNSC membership while Japan is also a member, and UNSC membership when Japan is not. For the former, we have but six country-year observations, which derive from the five countries – India, Nepal, Pakistan, Philippines, and South Korea – that have served together with Japan on the UNSC. Note that the UNSC terms for these five countries partially overlap with those of Japan's, such that Japan serves only one year out of its two-year term with each of them. The Philippines (1981), Nepal (1988), India (1972), and South Korea (1997) have each served with Japan on one occasion, whereas Pakistan (1976, 1993) has served with Japan during two different UNSC terms.³⁹ The dearth in the number of observations and the fact that each country serves with Japan for only a one-year window present us with a hard test case. We therefore consider the results suggestive at best.

We begin by presenting the descriptive evidence (Figure 3). We then analyze the same set of econometric regressions that we did above (Table 5).

Figure 3 shows that while the average share of ADB lending to countries not on the UNSC is about 7 percent, it is about 11 percent for UNSC members not serving with Japan on the UNSC, and nearly 16 percent for countries serving with Japan on the UNSC. For the subset of developing member-countries that served with Japan on the UNSC, the average share of ADB lending received is about 10 percent two years prior to service and 12 percent one year prior to service. The share is about 15 percent the year after service, and drops to less than 11 percent the

³⁹ India also served with Japan in 1972, but, at that time, India was not eligible for ADB loans due to Japanese concerns that India's size and the extent of its poverty would lead it to "consume too large a share of the institution's resources and, in a sense, dominate the institution (Kilby 2006)." See also Withol (1988).

year after. Figure 3 also makes clear that our evidence is driven by a few observations – our regression analysis is thus highly tentative at best.

Table 5 shows that the effect of UNSC membership with Japan is positive and statistically significant at the 0.05 level or better. The effect holds in the presence of year and region fixed effects, year and country fixed effects, and controlling for first-order autocorrelation of the error term with country and year fixed effects. Note that all these regressions include our set of control variables. The effect of UNSC membership without Japan is not robustly statistically-significant at conventional levels. We should note, however, that these results are sensitive to the presence of outliers. The prominent outlier that is dropped is South Korea 1997, where the East Asian Financial Crisis loomed large. If we trim the upper and lower bounds of the distribution of ADB lending, the significance switches - UNSC membership with Japan is no longer statistically significant, while UNSC membership without Japan is statistically significant. The latter is consistent with the robustness of the overall results presented in section 5.3 above. As for the loss of significance when we drop observations here, considering that we began this analysis with only six observations of UNSC members participating with Japan, we are not surprised that the result does not hold when we drop even one additional observation in the tail of the distribution.

Though history has yet to provide us with ample data, our results suggest that the effect of UNSC membership on ADB lending to developing member-countries may be influenced more when Japan is a member of the UNSC than when it is not. Considering that the United States is a permanent and consistently active member of the UNSC, we take this as preliminary evidence that political pressure on the ADB to assist UNSC members may come more systematically from Japan.



Figure 3: Allocation of ADB lending by UNSC membership – with and without Japan on the UNSC

Note: The horizontal line shows the mean share of ADB lending across the entire sample (0.07, n=467).

Table 5: The effect of UNSC	membership – wi	ith and without Jap	oan – on ADB lending
-----------------------------	-----------------	---------------------	----------------------

UNSC with Japan	(1) OLS 0.085***	(2) OLS 0.081***	(3) OLS 0.079***	(4) Fixed country effects 0.081***	(5) Fixed country effects with AR(1) 0.036**
1	(0.032)	(0.031)	(0.025)	(0.023)	(0.017)
UNSC without Japan	0.035** (0.017)	0.028** (0.016)	0.007 (0.013)	0.006 (0.012)	0.013 (0.011)
Japan aid share			0.332*** (0.064)	0.334*** (0.074)	0.169** (0.074)
US aid share			0.156*** (0.047)	0.050 (0.051)	-0.027 (0.050)
World exports			0.803*** (0.232)	0.537 (0.376)	-0.162 (0.413)
World imports			0.112 (0.213)	-0.106 (0.271)	0.264 (0.425)
Japan imports			-0.156 (0.111)	0.077 (0.141)	-0.225 (0.239)
US exports			-0.365** (0.151)	-0.889*** (0.250)	-0.026 (0.295)
Population			1.312*** (0.224)	4.670 (3.162)	8.225 (5.348)
Population-squared			3.903*** (0.591)	- 12.255*** (4.658)	- 15.304** (7.577)
GDP/capita			0.007*** (0.002)	-0.008** (0.003)	-0.007 (0.005)
Polity			-0.0002 (0.001)	-0.002*** (0.001)	-0.002* (0.001)
Constant	0.072*** (0.004)	0.119*** (0.034)	0.174*** (0.033)	0.071 (0.105)	-0.202** (0.079)
Asian-Region fixed effects	NO	YES	YES	NO	NO
Country fixed effects	NO	NO	NO	YES	YES
Year fixed effects	NO	YES	YES	YES	YES
Number of observations	467	467	467	467	443
Number of countries	n/a	n/a	n/a	24	23
(Adjusted) R2	0.019	0.164	0.491	0.163	0.201

6. Conclusion

Our main results show that when a developing country rises to political prominence by being elected to the UN Security Council, its share of loans from the Asian Development Bank increases by about two percentage points. As ADB projects take some time to be put together, the effect during the second year of the two-year UNSC term is stronger: share of ADB loans increases by three percentage points. This effect may be as high as four percentage points if a country happens to be serving when Japan is also a UNSC member. Considering that the average share of ADB lending for a typical country is about 7.5 percent, these effects are substantial – increasing a country's share by upwards of 25 percent of what it would obtain in the absence of UNSC membership. We take this as evidence that the major vote-holders at the ADB – mainly Japan – use their leverage over it to influence the UNSC's voting patterns.

It is difficult to know the effectiveness of this vote-buying strategy – we do not know what the counterfactual world with no political manipulation would look like. But we can at least report that Asian countries have never voted against Japan in the 274 resolutions passed during their period of concurrent service on the UNSC. Asian countries have abstained on a resolution supported by Japan only eight times, and voted for a resolution where Japan abstained twice. Larger countries Pakistan (1976, 1993) and India (1992) – which may be more difficult to influence – comprise these exceptions. The others – Philippines (1981), Nepal (1988), and South Korea (1997) – voted with Japan one hundred percent of the time. Interestingly, while these Asian countries collectively voted with Japan 96 percent of the time, on these same resolutions, they voted slightly less often with the United States (94 percent of the time). Of course, there is massive selection bias in what resolutions go forward at the UNSC, and, moreover, Asian

countries sometimes vote together out of shared regional interests – so this evidence should be taken as purely suggestive.

The United States has traditionally played the role of "system maker and privilege taker" – its role as creator, maintainer, and defender of the liberal economic order has allowed it to avoid costly economic adjustments at the expense of other countries (Mastanduno 2009). Most countries stuck with this arrangement because the United States was all-powerful; it took what it could, while they suffered what they had to. This implicit bargain may unravel is US relative power continues to decline. The recent proliferation of regional initiatives and the growing number of regional organizations reflect this shift in the locus of global governance. Regional organizational structures accord emerging market countries a greater and more representative share of voting power, and thus may possess greater legitimacy vis-à-vis their global counterparts (see, Lipscy 2006). They are generally perceived as better than global organizations in safeguarding developing country interests.

A move from global institutions to regional ones does not, however, circumvent the fundamental issue of political manipulation. It is likely that where global hegemons like the United States once reigned supreme within global organizations, regional hegemons will eventually rise to take their place in an increasingly multi-polar world, dominating regional organizations. This political manipulation may not be a bad thing. For, a world without politically manipulated international organizations might be a world without any organizations at all, and thus less international cooperation. Political manipulation may thus be something the world must live with in order to have forums for coordination to address issues that cross borders. And it may be appropriate to see the rise of regional hegemons taking their place as the principal manipulators in a multi-polar world. So, in closing, we emphasize the importance of

understanding that political manipulation is likely to continue even if global power becomes decentralized along regional lines. The change will be a shift in hegemons, from global to regional. It is, thus, no wonder that rising powers, notably China, are investing in regional organizations as a solution to what they perceive to be the failures of global alternatives.

References

- Abadie, Alberto, David Drukker, Jane Leber Herr, and Guido W. Imbens. 2004. Implementing Matching Estimators For Average Treatment Effects in Stata. *Stata Journal* 4 (3):290-311.
- Abbott, Kenneth and Duncan Snidal. 1998. Why States Act Through Formal International Organizations. *Journal of Conflict Resolution* 42 (1):3-32.
- Ayres, Ian and Robert Gerner. 1989. Filling Gaps in Incomplete Contracts: An Economic Theory of Default Rules. *The Yale Law Journal* 99 (1): 87-130.
- Bello, Warren. 2000. ADB 2000: Senior Officials and Internal Documents Paint Institution in Confusion. *Focus on Trade* No. 49.
- Bernauer, Thomas and Ruloff, D. (eds.). 1999. *The Politics of Positive Incentives in Arms Control.* Columbia, SC: University of South Carolina Press.
- Bhagwati, Jagdisgh. 1993. "Regionalism and Multilateralism: An Overview." in Jaime de Melo and Arvind Panagariya (eds), <u>New Dimensions in Regional Integration</u>. New York: Cambridge University Press.
- Birdsall, Nancy, Stijn Claessens, and Ishac Diwan. 2003. Policy Selectivity Forgone: Debt and Donor Behavior in Africa. *World Bank Economic Review* 17 (3): 409-435.
- Bueno de Mesquita, Bruce and Alastair Smith. 2010. The Pernicious Consequences of UN Security Council Membership. *The Journal of Conflict Resolution* 54 (3): forthcoming.

- Bulmer-Thomas, Victor. 1998. The Central American Common Market: From Closed to Open Regionalism. World Development 26 (2): 313-322.
- Camerer, Colin Samuel Issacharof, George Loewenstein, Ted O'Donoghue, and Matthew Rabin.
 2003. Regulation For Conservatives: Behavioral Economics And The Case For 'Asymmetric Paternalism'. University of Pennsylvania Law Review 151: 1211-1254.
- Chapman, Terrence, and Dan Reiter. 2004. The United Nations Security Council and the Rally 'Round the Flag' Effect. *Journal of Conflict Resolution* 48 (6): 886-909.
- Chapman, Terrence. 2007. International Security Institutions, Domestic Politics, and Institutional Legitimacy. *Journal of Conflict Resolution* 51 (1): 134-166.
- Chapman, Terrence. Forthcoming. *Securing Approval: Domestic Politics and Multilateral Authorization for War*. Chicago, Illinois: University of Chicago Press.
- Cheibub, Jose Antonio, Jennifer Gandhi, and James Raymond Vreeland. 2010. Democracy and Dictatorship Revisited. *Public Choice* 143 (1-2): 67-101.
- Davis, Christina L. 2009. Linkage Diplomacy: Economic and Security Bargaining in the Anglo-Japanese Alliance, 1092-23. *International Security* 33 (3): 143-179.
- Dehejia, Rajeev. 2005. Practical Propensity Score Matching: A Reply to Smith and Todd? Journal of Econometrics 125: 355-364.
- Devarajan, Shantayana, Andrew Sunil Rajkumar, and Vinaya Swaroop. 1999. What Does Aid to Africa Finance? *World Bank Policy Research Paper No. 2092*.

Dore, Ronald. 1997. Japan, Internationalism, and the UN. London: Routledge.

Dorussen, Han. 2001. Mixing Carrots with Sticks: Evaluating the Effectiveness of Positive Incentives. *Journal of Peace Research* 38 (2): 251-262.

- Dreher, Axel, Jan-Egbert Sturm, and James Raymond Vreeland. 2009a. Global Horse Trading: IMF Loans for Votes in the United Nations Security Council? *European Economic Review* 53 (7): 742-757.
- Dreher, Axel, Jan-Egbert Sturm, and James Raymond Vreeland. 2009b. Development Aid and International Politics: Does Membership on the UN Security Council Influence World Bank Decisions? *Journal of Development Economics* 88 (1): 1-18.
- Dreher, Axel, Jan-Egbert Sturm, and James Raymond Vreeland. 2010. Does membership on the UN Security Council influence IMF conditionality? Center for European, Governance and Economic Development Research Working Paper No. 104.
- Dreher, Axel and James Raymond Vreeland. 2010. Idiosyncratic Election to the UN Security Council. Ms.
- Drezner, Daniel W., 1999. The Sanctions Paradox: Economic Statecraft and International Relations. New York: Cambridge University Press.
- Drifte, Reinhard. 2000. Japan's Quest for a Permanent Security Council Seat: A Matter of Pride or Justice? New York: St. Martin's Press.
- Edstrom, Bert. 1999. *Evolving Foreign Policy Doctrine: From Yoshida to Miyazawa*. New York: St. Martin's Press.
- Eichengreen, Barry. 2003. What To Do With The Chiang Mai Initiative. Asian Economic Papers. 2 (1): 1-49.
- Erfat, Asif. 2010. Toward Internationally Regulated Goods: Controlling the Trade in Small Arms and Light Weapons. *International Organization* 64 (1): 91-131.
- Fang, Songying, 2008, The informational role of international institutions and domestic politics, *American Journal of Political Science* 52, 2: 304-321.

- Fleck, Robert K and Christopher Kilby. 2006. World Bank Independence: A Model and Statistical Analysis of U.S. Influence. *Review of Development Economics* 10 (2): 224-240.
- Frankel, Rafael D. and Rebecca Davis Gibbons. 2010. Aid, Influence and the United Nations: Testing the Japanese Case. Georgetown University Government Department, Ms.
- Gowa, Joanne and Edward Mansfield. 1993. Power Politics and International Trade. *American Political Science Review*. 87 (2): 408-420.
- Grimes, William. 2010a. The Future of Regional Liquidity Arrangements in East Asia: Lessons from the Global Financial Crisis. Ms. Boston University.
- Grimes, William. 2010b. The Asian Monetary Fund Reborn? Implications of Chiang Mai Initiative Multilateralization. Ms. Boston University.
- Heckman, James, Hidehiko Ichimura, and Petra Todd. 1998. Matching as an Econometric Evaluation Estimator. *Review of Economic Studies*. 65 (2): 261-294.
- Henning, Randall C. 2009. The Future Of The Chiang Mai Initiative: An Asian Monetary Fund? Peterson Institute for International Economics Policy Brief. PB09-5: 1-9.
- Hook, Glenn D, Julie Gibson, Christopher W. Hughes, and Hugo Dobson. 2001. Japan's International Relations: Politics, Economics, and Security. New York: Routledge.

Heenan, Patrick. 1998. The Japan Handbook. Chicago, Illinois: Fitzroy Dearborn Publishers.

- Heldt, Birger and Peter Wallensteen. 2006. *Peacekeeping Operations: Global Patterns of Intervention and Success, 1948-2004.* 2nd ed. Folke Bernadotte Academy Publications.
- Heston, Alan, Robert Summers, and Bettina Aten. 2002. Penn World Table Version 6.1. Center for International Comparisons at the University of Pennsylvania (CICUP), Philadelphia, Pa.

- Ho, Daniel E, Imai Kosuke, Gary King, and Elizabeth A. Stuart. 2007. Matching as Nonparametric Preprocessing for Reducing Model Dependence in Parametric Causal Inference. *Political Analysis*. 15 (3): 199-236.
- Hopkins, Daniel, and Beth Simmons. 2005. The Constraining Power of International Treaties: Theory and Methods. *The American Political Science Review*. 99 (4): 623-631.
- Hurd, Ian. 2007. After Anarchy: Legitimacy and Power in the United Nations Security Council.Princeton, New Jersey: Princeton University Press.
- Hurd, Ian and Bruce Cronin. 2008. *The UN Security Council and the Legitimacy of International Authority* (eds). New York: Routledge.
- Imai, Kosuke and David A. van Dyk. 2004. Causal Inference with General Treatment Regimes:Generalizing the Propensity Score. *Journal of the American Statistical Association* 99 (467):854-866.
- International Monetary Fund. 2004. *Direction of Trade on CD-ROM*. International Monetary Fund: Washington, DC.
- Johnson, Eric and Daniel Goldstein. 2003. Do Defaults Save Lives? Science 302: 1338-1339.
- Kappagoda, Nihal. 1995. *The Asian Development Bank*. Boulder, Colorado: Lynne Rienner Publishers.
- Kilby, Christopher. 2006. Donor Influence in Multilateral Development Banks: The Case of the Asian Development Bank. *Review of International Organizations* 1 (2): 173-195.
- Kilby, Christopher. 2009. The Political Economy of Conditionality: An Empirical Analysis of World Bank Loan Disbursements. *Journal of Development Economics* 89 (1): 51-61.

- Kilby, Christopher. 2010. Informal Influence in the Asian Development Bank. Paper prepared for The Political Economy of International Financial Institutions—Egon Sohmen Memorial Conference, Tübingen, Germany, June 10-13.
- Korobkin, Russell. 1998. The Status Quo Bias and Contract Default Rules. *Cornell Law Review* 83: 608-687.
- Kuziemko, Ilyana and Eric Werker. 2006. How Much Is a Seat on the Security Council Worth?Foreign Aid and Bribery at the United Nations. *Journal of Political Economy* 114 (5): 905-930.
- Leuven, Edwin and Barbara Sianesi. 2003. PSMATCH2: Stata module to perform full Mahalanobis and propensity score matching, common support graphing, and covariate imbalance testing. Available: http://ideas.repec.org/c/boc/bocode/s432001.html. This version 3.1.5 2 (May 2009).
- Lim, Daniel Yew Mao. 2010 Reconceptualizing International Law In International Relations Theory: An Agent-Centered Network Approach. Georgetown University School of Foreign Service Ms.
- Lipscy, Phillip. 2003. Japan's Asian Monetary Fund Proposal. *Stanford Journal of East Asian Affairs* 3 (1): 93-104.
- Malone, David. 2004. *The UN Security Council: From the Cold War to the 21st Century*. Oxford: Oxford University Press.
- Mansfield, Edward D and Helen V. Milner. 1999. The New Wave of Regionalism. *International Organization* 53 (3): 589-627.
- Marchesi, Silvia. 2003. Adoption of an IMF Programme and Debt Rescheduling. An Empirical Analysis. *Journal of Development Economics* 70 (2): 403-423.

- Marchesi, Silvia and Alessandro Missale. 2004. What Does Aid Motivate Lending and Aid to the HIPCs? *Centro Studi Luca D'Agliano Development Studies Working Papers No. 189*.
- Marshall, Monty and Keith Jaggers. 2000. Polity IV Project. Polity IV dataset. Center for International Development and Conflict Management, University of Maryland, College Park, MD.
- Mastanduno, Michael. 2009. System Maker and Privilege Taker: US Power and the International Political Economy. *World Politics*. 61 (1): 121-154.
- Milner, Helen and Dustin Tingley. 2010a. The Domestic Politics of Foreign Aid: American Legislators and the Politics of Donor Countries. *Economics and Politics* (forthcoming).
- Milner, Helen and Dustin Tingley I. 2010b. The Choice of Multilateralism: Foreign Aid and American Foreign Policy. *Mimeo*.
- Moravcsik, Andrew. 1997. Taking Preferences Seriously: A Liberal Theory of International Politics. *International Organization* 51 (4): 513-553.
- Morgenthau, Hans. 1962. A Political Theory of Foreign Aid. *The American Political Science Review* 56: 301-309.
- Nichols, Austin. 2007. Causal Inference With Observational Data. Stata Journal 7 (4): 507-541.
- OECD Development Assistance Committee. 2004. International Development Statistics on CD-ROM, OECD: Paris.
- O'Neill, Barry. 1985. Power and Satisfaction in the United Nations Security Council. Weltwirtschaftliches Archiv 121 (3): 438-447.
- Pomfret, Richard. 2006. Is Regionalism an Increasing Feature of the World Economy? *The World Economy* 30 (6): 923-947.

- Rosenbaum, Paul and Donald Rubin. 1983. The Central Role of the Propensity Score in Observational Studies for Causal Effects. *Biometrika* 70: 41-55.
- Rozman, Gilbert. 2007. "Japanese Strategic Thinking on Regionalism." in Gilbert Rozman,
 Kazuhiko Togo, and Joseph P. Ferguson (eds), <u>Japanese Strategic Thought Toward Asia</u>.
 New York: Palgrave McMillian.

Russett, Bruce. 1997. The Once and Future Security Council. St. Martin's Press: New York.

- Saiegh, Sebastian. 2011. Ruling by Statute: How Uncertainty and Vote-Buying shape Lawmaking. New York: Cambridge University Press.
- Severino, Rodolfo C. 2007. "International Law, International Relations and Compliance." in Gennady Chufrin and Mark Hong (eds), <u>Russia-ASEAN Relations: New Directions</u>. Singapore: ISEAS Publications.
- Smith, Jeffrey A. and Petra E. Todd. 2005. Does Matching overcome LaLonde's Critique of Nonexperimental Estimators? *Journal of Econometrics* 125: 305-353.
- Stone, Randall W. 2011. Controlling Institutions: International Organizations and the Global Economy. New York: Cambridge University Press.
- Strand, Jonathan. 2011. *Regional Development Banks: Lending with a Regional Flavor*. New York: Routledge.
- Strand, Jonathan R. and David P. Rapkin. 2010. Weighted Voting in the United Nations Security Council: A Simulation. *Simulation & Gaming* doi:10.1177/1046878110365514.
- Thaler, Richard H. and Cass R. Sunstein. 2008. Nudge: Improving Decisions About Health, Wealth, and Happiness. New Haven: Yale University Press.

Thacker, Strom C. 1999. The High Politics of IMF Lending. World Politics 52: 38-74.

- Thompson, A. 2006. Coercion Through IOs: The Security Council and the Logic of Information Transmission. *International Organization* 60 (1): 1-34.
- Tsebelis, George. 1994. The Power of the European Parliament as a Conditional Agenda Setter. *The American Political Science Review* 88 (1): 128-142.
- Tsebelis, George and Amie Kreppel. 1998. The History of Conditional Agenda Setting in European Institutions. *European Journal of Political Research* 33: 41-71.
- Tuman, John P., Jonathan R. Strand, and Craig F. Emmert. 2009. The Disbursement Pattern of Japanese Foreign Aid: A Reappraisal. *Journal of East Asian Studies* 9 (2):219-248.
- Vaubel, Roland. 1986. A Public Choice Approach to International Organization. *Public Choice* 51: 39-57.
- Voeten, Erik. 2001. Outside Options and the Logic of Security Council Action. *American Political Science Review* 95 (4): 845-858.
- Voeten, Erik. 2005. The Political Origins of the UN Security Council's Ability to Legitimize the Use of Force. *International Organization* 59 (3): 527-557.
- Volden C, Carrubba CJ, 2004. The Formation of Oversized Coalitions in Parliamentary Democracies. *American Journal of Political Science* 48;3; 521-537.
- Weiss, Jessica. 2008. The 2005 Anti-Japanese Protests in China and UNSC Reform. Princeton University Woodrow Wilson School, Ms.
- Withol, Robert. 1988. *The Asian Development Bank and Rural Development: Policy and Practice*. New York: St Martin's Press.
- World Bank. 2004. World Development Indicators on CD-ROM. World Bank: Washington, DC.

Appendix: Descriptive Statistics

		Std.					# of
Variable	Mean	Dev.	Min	Max	Unit	Original source	obs.
ADB share	0.075	0.080	0.0001	0.52	Share	OECD Development Assistance Committee (2004)*	467
UNSC	0.064	0.245	0	1	Binary	http://www.un.org/sc/members.asp	467
UNSC-Japan	0.013	0.113	0	1	Binary	http://www.un.org/sc/members.asp	467
UNSC-No Japan	0.051	0.221	0	1	Binary	http://www.un.org/sc/members.asp	467
Population (t-1)	0.044	0.094	0.0002	0.43	Share	Heston et al. (2002), World Bank (2004)*	467
Peacekeeping troops	2.094	2.910	0	8.98	Log	Heldt and Wallensteen (2006)	455
Japan aid share	0.067	0.074	0	0.34	Share	OECD Development Assistance Committee (2004)*	467
US aid share	0.064	0.079	0	0.39	Share	OECD Development Assistance Committee (2004)*	467
World exports (t-2)	0.053	0.058	0	0.33	Share	International Monetary Fund (2004)*	467
World imports (t-2)	0.053	0.066	0	0.41	Share	International Monetary Fund (2004)*	467
Japan imports (t-2)	0.059	0.086	0	0.44	Share	International Monetary Fund (2004)*	467
US exports (t-2)	0.051	0.065	0	0.30	Share	International Monetary Fund (2004)*	467
GDP/capita (t-1)	3.262	2.606	0.41	14.79	\$000 PPP 1996	Heston et al. (2002), World Bank (2004)*	467
Polity (t-1)	0.698	6.509	-9	10	-10 to 10	Marshall and Jaggers (2000)*	467
East Asia	0.131	0.337	0	1	Binary	http://www.adb.org/Countries/default.asp	467
Central & Western Asia	0.113	0.318	0	1	Binary	http://www.adb.org/Countries/default.asp	467
Pacific Islands	0.113	0.318	0	1	Binary	http://www.adb.org/Countries/default.asp	467
Southeast Asia	0.396	0.490	0	1	Binary	http://www.adb.org/Countries/default.asp	467
South Asia	0.246	0.431	0	1	Binary	http://www.adb.org/Countries/default.asp	467
	-				-	-	

*Taken from Kilby (2006)